Author's response to reviews

Title: A process-based framework to guide Nurse Practitioners integration into primary healthcare teams: results from a logic analysis.

Authors:

Damien Contandriopoulos (Damien.contandriopoulos@umontreal.ca)
Astrid Brousselle (Astrid.Brousselle@usherbrooke.ca)
Carl-Arty Dubois (carl.ardy.dubois@umontreal.ca)
Mélanie Perroux (perrouxmelanie@gmail.com)
Marie-Dominique Beaulieu (marie-dominique.beaulieu@umontreal.ca)
Isabelle Brault (isabelle.brault@umontreal.ca)
Kelley Kilpatrick (kelley.kilpatrick@umontreal.ca)
Danielle D'Amour (danielle.damour@umontreal.ca)
Esther Sansgter-Gormley (egorm@uvic.ca)

Version: 5 Date: 21 January 2015

Author's response to reviews: see over
January 20, 2015

Dr. May Nawal Lutfiyya
Associate Editor
BMC Health Services Research

MS: 1674317641345608 Research article
A process-based framework to guide NP integration into primary healthcare teams. Damien Contandriopoulos, Astrid Brousselle, Carl-Ardy Dubois, Mélanie Perroux, Marie-Dominique Beaulieu, Isabelle Brault, Kelley Kilpatrick, Danielle D'Amour and Esther Sansgter-Gormley
BMC Health Services Research

Dear Professor Lutfiyya:

As requested, we made the formatting changes concerning the title page, the abstract and the references. We added as well:

- the full names and affiliations of all the ethics committees that approved the study
- the competing interest section
- the authors' contributions
- the acknowledgements

We hope the paper in its current version will be considered acceptable for publication.

Sincerely,